THE ROYAL SOCIETY PUBLISHING

PROCEEDINGS B

Long-lasting vocal plasticity in adult marmoset monkeys

Lingyun Zhao, Bahar Boroumand Rad and Xiaogin wang

Article citation details Proc. R. Soc. B 286: 20190817.

http://dx.doi.org/10.1098/rspb.2019.0817

Review timeline

Original submission: 1st revised submission: 6 April 2019 2nd revised submission: 3 June 2019 Final acceptance:

28 January 2019 4 June 2019

Note: Reports are unedited and appear as submitted by the referee. The review history appears in chronological order.

Review History

RSPB-2019-0205.R0 (Original submission)

Review form: Reviewer 1

Recommendation

Accept with minor revision (please list in comments)

Scientific importance: Is the manuscript an original and important contribution to its field? Good

General interest: Is the paper of sufficient general interest? Excellent

Quality of the paper: Is the overall quality of the paper suitable? Good

Is the length of the paper justified? Yes

Should the paper be seen by a specialist statistical reviewer? No

Reports © 2019 The Reviewers; Decision Letters © 2019 The Reviewers and Editors; Responses © 2019 The Reviewers, Editors and Authors. Published by the Royal Society under the terms of the Creative Commons Attribution License http://creativecommons.org/licenses/by/4.0/, which permits unrestricted use, provided the original author and source are credited

Do you have any concerns about statistical analyses in this paper? If so, please specify them explicitly in your report.

No

It is a condition of publication that authors make their supporting data, code and materials available - either as supplementary material or hosted in an external repository. Please rate, if applicable, the supporting data on the following criteria.

Is it accessible? No Is it clear? N/A Is it adequate? N/A

Do you have any ethical concerns with this paper? No

Comments to the Author

One issue which mars this ms is a number of overstatements. Here for the Abstract: "Non-human primates, however, have long been thought to have little plasticity in vocal production, especially in adulthood, which would limit their ability in carrying out effective vocal communication in natural environment". Given the authors note that developing marmosets can show vocal plasticity, and the Introduction reviews many articles (refs 17-19, 23-32) showing adult plasticity, the first part is an overstatement and the second part is not logical since the absence of vocal plasticity in other species is not correlated with an inability to carry out effective communication which must be viewed in any evolutionary sense as environmental niche-appropriate communication.

What distinguishes this study is NOT a novel demonstration of adult vocal plasticity in marmosets, but the precise quantification (c.f lines 78-80) of such plasticity under their lab-based test and observational conditions. In this context, the claim that the ability of adult marmosets to systematically alter vocalizations spectrotemporal structures during vocal interactions with conspecifics in a manner dependent on acoustic context and experience is crucial to demonstrate vocal plasticity and learning, is again exaggerated - it is a strong and very important manifestation of vocal plasticity and learning but the basic demonstration of a capacity for vocal plasticity and learning is already present in the studies reviewed.

Thus the claim at the end of the Introduction that "Our results provide new evidence for the voluntary control of spectrotemporal structures of vocalizations by adult marmosets and suggest the possibility of context related learning in the vocal behaviors of this species." is exaggerated. This study provides further evidence for the voluntary control of spectrotemporal structures of vocalizations by adult marmosets and is consistent with evidence for context related learning in the vocal behaviors of this species. This doesn't devalue this study which is a very well controlled exercise that gives confidence in such vocal learning but it does mean the authors must mute the overstatements.

Minor errors:

English expression can be improved. Here are some early examples:line 27 environments; 129

contexts; l30 of instead of on; similarly in the Discussion (note this is in the context that this was a well-written paper otherwise)

Fig 1A, the call shown in the spectrograms must be identified in the legend so the figure is standalone.

Review form: Reviewer 2

Recommendation Major revision is needed (please make suggestions in comments)

Scientific importance: Is the manuscript an original and important contribution to its field? Excellent

General interest: Is the paper of sufficient general interest? Good

Quality of the paper: Is the overall quality of the paper suitable? Good

Is the length of the paper justified? Yes

Should the paper be seen by a specialist statistical reviewer? Yes

Do you have any concerns about statistical analyses in this paper? If so, please specify them explicitly in your report. No

It is a condition of publication that authors make their supporting data, code and materials available - either as supplementary material or hosted in an external repository. Please rate, if applicable, the supporting data on the following criteria.

Is it accessible? N/A Is it clear? N/A Is it adequate? N/A

Do you have any ethical concerns with this paper? No

Comments to the Author

A fascinating and well-written read! However, authors should address the following before publication.

1. Describe and include any research on humans that parallels specifically the interfering talker

Marmoset scenario. The Lombard effect and other effects included are not adequate.

2. Define "antiphonal calling" for the reader.

3. Be more specific in defining low and high frequency noise --- does this mimic an interfering call? If not, then what? What specific environmental noises are also being simulated. How accurate are the simulated sounds in reproducing interfering calls and true environemental noises?

4. The perturbation signal occurs "after the onset of a phee call". Is it overlapping the call and to what extent? Would there be a different effect if not overlapping?

5. Please describe how the virtual calls and the perturbations are actually implemented with the Marmoset calls in real time. The authors say nothing about this!

6. The statistical analysis appears careful but needs further clarity.

7. In a real dialogue, would a receiving marmoset not get confused if the caller is anticipating a shift not present and thus would the learning effect stop soon?

8. In the discussion, authors should tone down the generalizations based on the limited experimental setup.

9. Figures 1,2 and parts of 3,4 are very confusing in terms of timing of the various signal components. Please clarify.

4.

Decision letter (RSPB-2019-0205.R0)

06-Mar-2019

Dear Dr Wang:

I am writing to inform you that your manuscript RSPB-2019-0205 entitled "Long-lasting Vocal Plasticity in Adult Marmoset Monkeys" has, in its current form, been rejected for publication in Proceedings B.

This action has been taken on the advice of referees, who have recommended that substantial revisions are necessary. With this in mind we would be happy to consider a resubmission, provided the comments of the referees are fully addressed. However please note that this is not a provisional acceptance. To add to the referees' comments, please discuss the social experience that your subjects had and consider how these results may reflect their behavior in a more natural context, and carefully reframe the manuscript to ensure that you do not overstate your findings.

The resubmission will be treated as a new manuscript. However, we will approach the same reviewers if they are available and it is deemed appropriate to do so by the Editor. Please note that resubmissions must be submitted within six months of the date of this email. In exceptional circumstances, extensions may be possible if agreed with the Editorial Office. Manuscripts submitted after this date will be automatically rejected.

Please find below the comments made by the referees, not including confidential reports to the Editor, which I hope you will find useful. If you do choose to resubmit your manuscript, please upload the following:

1) A 'response to referees' document including details of how you have responded to the comments, and the adjustments you have made.

2) A clean copy of the manuscript and one with 'tracked changes' indicating your 'response to referees' comments document.

3) Line numbers in your main document.

To upload a resubmitted manuscript, log into http://mc.manuscriptcentral.com/prsb and enter your Author Centre, where you will find your manuscript title listed under "Manuscripts with Decisions." Under "Actions," click on "Create a Resubmission." Please be sure to indicate in your cover letter that it is a resubmission, and supply the previous reference number.

Sincerely,

Prof Sarah F. Brosnan Editor, Proceedings B mailto: proceedingsb@royalsociety.org

Associate Editor

Comments to Author:

Your manuscript has now been evaluated by two experts in the field. Although they both found a lot to like about your study, they have raised some concerns. I am therefore recommending rejection of the manuscript in its present form. Having said that, I would also encourage you to re-submit a revised manuscript. In your revision, it would be useful if you could make sure that the statistical analyses are made as clear and as accessible as possible.

Reviewer(s)' Comments to Author:

Referee: 1

Comments to the Author(s)

One issue which mars this ms is a number of overstatements. Here for the Abstract: "Non-human primates, however, have long been thought to have little plasticity in vocal production, especially in adulthood, which would limit their ability in carrying out effective vocal communication in natural environment". Given the authors note that developing marmosets can show vocal plasticity, and the Introduction reviews many articles (refs 17-19, 23-32) showing adult plasticity, the first part is an overstatement and the second part is not logical since the absence of vocal plasticity in other species is not correlated with an inability to carry out effective communication which must be viewed in any evolutionary sense as environmental niche-appropriate communication.

What distinguishes this study is NOT a novel demonstration of adult vocal plasticity in marmosets, but the precise quantification (c.f lines 78-80) of such plasticity under their lab-based test and observational conditions. In this context, the claim that the ability of adult marmosets to systematically alter vocalizations spectrotemporal structures during vocal interactions with conspecifics in a manner dependent on acoustic context and experience is crucial to demonstrate vocal plasticity and learning, is again exaggerated - it is a strong and very important manifestation of vocal plasticity and learning but the basic demonstration of a capacity for vocal plasticity and learning is already present in the studies reviewed.

Thus the claim at the end of the Introduction that "Our results provide new evidence for the voluntary control of spectrotemporal structures of vocalizations by adult marmosets and suggest the possibility of context related learning in the vocal behaviors of this species." is exaggerated. This study provides further evidence for the voluntary control of spectrotemporal structures of vocalizations by adult marmosets and is consistent with evidence for context related learning in the vocal behaviors of this species. This doesn't devalue this study which is a very well controlled

exercise that gives confidence in such vocal learning but it does mean the authors must mute the overstatements.

Minor errors:

English expression can be improved. Here are some early examples:line 27 environments; l29 contexts; l30 of instead of on; similarly in the Discussion (note this is in the context that this was a well-written paper otherwise)

Fig 1A, the call shown in the spectrograms must be identified in the legend so the figure is standalone.

Referee: 2

Comments to the Author(s)

A fascinating and well-written read! However, authors should address the following before publication.

1. Describe and include any research on humans that parallels specifically the interfering talker Marmoset scenario. The Lombard effect and other effects included are not adequate.

2. Define "antiphonal calling" for the reader.

3. Be more specific in defining low and high frequency noise --- does this mimic an interfering call? If not, then what? What specific environmental noises are also being simulated. How accurate are the simulated sounds in reproducing interfering calls and true environemental noises?

4. The perturbation signal occurs "after the onset of a phee call". Is it overlapping the call and to what extent? Would there be a different effect if not overlapping?

5. Please describe how the virtual calls and the perturbations are actually implemented with the Marmoset calls in real time. The authors say nothing about this!

6. The statistical analysis appears careful but needs further clarity.

7. In a real dialogue, would a receiving marmoset not get confused if the caller is anticipating a shift not present and thus would the learning effect stop soon?

8. In the discussion, authors should tone down the generalizations based on the limited experimental setup.

9. Figures 1,2 and parts of 3,4 are very confusing in terms of timing of the various signal components. Please clarify.

4.

Author's Response to Decision Letter for (RSPB-2019-0205.R0)

See Appendix A.

RSPB-2019-0817.R0

Review form: Reviewer 2

Recommendation Accept as is

Scientific importance: Is the manuscript an original and important contribution to its field? Good

General interest: Is the paper of sufficient general interest? Acceptable

Quality of the paper: Is the overall quality of the paper suitable? Good

Is the length of the paper justified? Yes

Should the paper be seen by a specialist statistical reviewer? No

Do you have any concerns about statistical analyses in this paper? If so, please specify them explicitly in your report. No

It is a condition of publication that authors make their supporting data, code and materials available - either as supplementary material or hosted in an external repository. Please rate, if applicable, the supporting data on the following criteria.

Is it accessible? No Is it clear? N/A Is it adequate? N/A have any othical

Do you have any ethical concerns with this paper? No

Comments to the Author

Authors have satisfactorily addressed reviewer concerns.

Decision letter (RSPB-2019-0817.R0)

28-May-2019

Dear Dr wang

I am pleased to inform you that your manuscript RSPB-2019-0817 entitled "Long-lasting Vocal Plasticity in Adult Marmoset Monkeys" has been accepted for publication in Proceedings B.

The referee(s) have recommended publication, but also suggest some minor revisions to your manuscript. Therefore, I invite you to respond to the referee(s)' comments and revise your manuscript. Because the schedule for publication is very tight, it is a condition of publication that you submit the revised version of your manuscript within 7 days. If you do not think you will be able to meet this date please let us know.

To revise your manuscript, log into https://mc.manuscriptcentral.com/prsb and enter your Author Centre, where you will find your manuscript title listed under "Manuscripts with Decisions." Under "Actions," click on "Create a Revision." Your manuscript number has been appended to denote a revision. You will be unable to make your revisions on the originally submitted version of the manuscript. Instead, revise your manuscript and upload a new version through your Author Centre.

When submitting your revised manuscript, you will be able to respond to the comments made by the referee(s) and upload a file "Response to Referees". You can use this to document any changes you make to the original manuscript. We require a copy of the manuscript with revisions made since the previous version marked as 'tracked changes' to be included in the 'response to referees' document.

Before uploading your revised files please make sure that you have:

1) A text file of the manuscript (doc, txt, rtf or tex), including the references, tables (including captions) and figure captions. Please remove any tracked changes from the text before submission. PDF files are not an accepted format for the "Main Document".

2) A separate electronic file of each figure (tiff, EPS or print-quality PDF preferred). The format should be produced directly from original creation package, or original software format. PowerPoint files are not accepted.

3) Electronic supplementary material: this should be contained in a separate file and where possible, all ESM should be combined into a single file. All supplementary materials accompanying an accepted article will be treated as in their final form. They will be published alongside the paper on the journal website and posted on the online figshare repository. Files on figshare will be made available approximately one week before the accompanying article so that the supplementary material can be attributed a unique DOI.

Online supplementary material will also carry the title and description provided during submission, so please ensure these are accurate and informative. Note that the Royal Society will not edit or typeset supplementary material and it will be hosted as provided. Please ensure that the supplementary material includes the paper details (authors, title, journal name, article DOI). Your article DOI will be 10.1098/rspb.[paper ID in form xxxx.xxxx e.g. 10.1098/rspb.2016.0049].

4) A media summary: a short non-technical summary (up to 100 words) of the key findings/importance of your manuscript.

5) Data accessibility section and data citation

It is a condition of publication that data supporting your paper are made available either in the electronic supplementary material or through an appropriate repository.

In order to ensure effective and robust dissemination and appropriate credit to authors the dataset(s) used should be fully cited. To ensure archived data are available to readers, authors should include a 'data accessibility' section immediately after the acknowledgements section. This should list the database and accession number for all data from the article that has been made publicly available, for instance:

• DNA sequences: Genbank accessions F234391-F234402

- Phylogenetic data: TreeBASE accession number S9123
- Final DNA sequence assembly uploaded as online supplemental material
- Climate data and MaxEnt input files: Dryad doi:10.5521/dryad.12311

NB. From April 1 2013, peer reviewed articles based on research funded wholly or partly by RCUK must include, if applicable, a statement on how the underlying research materials – such as data, samples or models – can be accessed. This statement should be included in the data accessibility section.

If you wish to submit your data to Dryad (http://datadryad.org/) and have not already done so you can submit your data via this link

http://datadryad.org/submit?journalID=RSPB&manu=(Document not available) which will take you to your unique entry in the Dryad repository. If you have already submitted your data to dryad you can make any necessary revisions to your dataset by following the above link. Please see https://royalsociety.org/journals/ethics-policies/data-sharing-mining/ for more details.

6) For more information on our Licence to Publish, Open Access, Cover images and Media summaries, please visit https://royalsociety.org/journals/authors/author-guidelines/.

Once again, thank you for submitting your manuscript to Proceedings B and I look forward to receiving your revision. If you have any questions at all, please do not hesitate to get in touch.

Sincerely,

Prof Sarah F Brosnan Editor, Proceedings B mailto: proceedingsb@royalsociety.org

Associate Editor Board Member Comments to Author: I am pleased to recommend acceptance of your fascinating manuscript. Congratulations.

Reviewer(s)' Comments to Author:

Referee: 2

Comments to the Author(s). Authors have satisfactorily addressed reviewer concerns.

Decision letter (RSPB-2019-0817.R1)

04-Jun-2019

Dear Dr wang

I am pleased to inform you that your manuscript entitled "Long-lasting Vocal Plasticity in Adult Marmoset Monkeys" has been accepted for publication in Proceedings B.

You can expect to receive a proof of your article from our Production office in due course, please check your spam filter if you do not receive it. PLEASE NOTE: you will be given the exact page length of your paper which may be different from the estimation from Editorial and you may be asked to reduce your paper if it goes over the 10 page limit.

If you are likely to be away from e-mail contact please let us know. Due to rapid publication and an extremely tight schedule, if comments are not received, we may publish the paper as it stands.

If you have any queries regarding the production of your final article or the publication date please contact procb_proofs@royalsociety.org

Your article has been estimated as being 10 pages long. Our Production Office will be able to confirm the exact length at proof stage.

Open Access

You are invited to opt for Open Access, making your freely available to all as soon as it is ready for publication under a CCBY licence. Our article processing charge for Open Access is £1700. Corresponding authors from member institutions

(http://royalsocietypublishing.org/site/librarians/allmembers.xhtml) receive a 25% discount to these charges. For more information please visit http://royalsocietypublishing.org/open-access.

Paper charges

An e-mail request for payment of any related charges will be sent out shortly. The preferred payment method is by credit card; however, other payment options are available.

Electronic supplementary material:

All supplementary materials accompanying an accepted article will be treated as in their final form. They will be published alongside the paper on the journal website and posted on the online figshare repository. Files on figshare will be made available approximately one week before the accompanying article so that the supplementary material can be attributed a unique DOI.

Thank you for your fine contribution. On behalf of the Editors of the Proceedings B, we look forward to your continued contributions to the Journal.

Sincerely,

Editor, Proceedings B mailto: proceedingsb@royalsociety.org

Appendix A

Responses to Referees' Comments (RSPB-2019-0205)

We thank the two reviewers for their constructive and helpful comments. In the revised manuscript, we have carefully addressed all concerns raised by the reviewers. In addition to the specific changes in response to reviewers' comments, we have added more detailed clarifications in the figure legend. We hope the reviewers find our responses satisfactory and we'd be happy to address any further questions.

The following is a point-by-point list of all changes made in response to the comments by the two reviewers. The page and line numbers refer to those of the revised manuscript (please see the document marked with "tracked changes"). The reviewers' comments are quoted and marked by blue font, followed by our response in black font.

Associate Editor

Comments to Author:

Your manuscript has now been evaluated by two experts in the field. Although they both found a lot to like about your study, they have raised some concerns. I am therefore recommending rejection of the manuscript in its present form. Having said that, I would also encourage you to re-submit a revised manuscript. In your revision, it would be useful if you could make sure that the statistical analyses are made as clear and as accessible as possible.

We have expanded our report for statistical analysis in the main text according to the suggestion by Reviewer 2.

Reviewer(s)' Comments to Author:

Referee: 1

Comments to the Author(s)

One issue which mars this ms is a number of overstatements. Here for the Abstract: "Non-human primates, however, have long been thought to have little plasticity in vocal production, especially in adulthood, which would limit their ability in carrying out effective vocal communication in natural environment". Given the authors note that developing marmosets can show vocal plasticity, and the Introduction reviews many articles (refs 17-19, 23-32) showing adult plasticity, the first part is an overstatement and the second part is not logical since the absence of vocal plasticity in other species is not correlated with an inability to carry out effective communication which must be viewed in any evolutionary sense as environmental niche-appropriate communication.

We appreciate the reviewer's recognition of the importance of the results presented in this manuscript and understand the reviewer's concerns of overstatements. In light of the reviewer's critiques, we have carefully re-worded a number of statements throughout the manuscript to

make sure we do not overstate the findings of this study. The above-mentioned statement in the abstract has been revised as follows (line 24-28).

"Humans exhibit a high level of vocal plasticity in speech production which allows us to acquire both native and foreign languages, dialects and adapt to local accents in social communication. In comparison, non-human primates exhibit limited vocal plasticity, especially in adulthood, which would limit their ability to adapt to different social and environmental contexts in vocal communication."

What distinguishes this study is NOT a novel demonstration of adult vocal plasticity in marmosets, but the precise quantification (c.f lines 78-80) of such plasticity under their lab-based test and observational conditions. In this context, the claim that the ability of adult marmosets to systematically alter vocalizations spectrotemporal structures during vocal interactions with conspecifics in a manner dependent on acoustic context and experience is crucial to demonstrate vocal plasticity and learning, is again exaggerated - it is a strong and very important manifestation of vocal plasticity and learning but the basic demonstration of a capacity for vocal plasticity and learning is already present in the studies reviewed.

We thank the reviewer for your comment! We have revised the above-mentioned statement in the introduction as follows (line 96-99).

"While findings from field studies and behavioral observations have suggested a capacity of vocal learning and plasticity in adult marmosets, there is a lack of quantitative measurement to describe the degree of vocal plasticity and the extent of long-term plasticity in vocal production which is crucial to fully substantiate vocal plasticity and learning in adult marmosets."

Thus the claim at the end of the Introduction that "Our results provide new evidence for the voluntary control of spectrotemporal structures of vocalizations by adult marmosets and suggest the possibility of context related learning in the vocal behaviors of this species." is exaggerated. This study provides further evidence for the voluntary control of spectrotemporal structures of vocalizations by adult marmosets and is consistent with evidence for context related learning in the vocal behaviors of this species. This doesn't devalue this study which is a very well controlled exercise that gives confidence in such vocal learning but it does mean the authors must mute the overstatements.

We have revised the above-mentioned statement as follows (line 120-123).

"Our results provide further evidence for the voluntary control of spectrotemporal structures of vocalizations by adult marmosets and suggest specific types of external cues that may lead to context related learning in the vocal behaviors of this species."

Minor errors:

English expression can be improved. Here are some early examples:line 27 environments; 129 contexts; 130 of instead of on; similarly in the Discussion (note this is in the context that this was a well-written paper otherwise)

Thanks for pointing out these errors! We have corrected these errors and carefully checked the entire manuscript again.

Fig 1A, the call shown in the spectrograms must be identified in the legend so the figure is standalone.

We have labeled the example calls in Fig 1A and added detailed explanations in the figure legend.

Referee: 2

Comments to the Author(s) A fascinating and well-written read! However, authors should address the following before publication.

1. Describe and include any research on humans that parallels specifically the interfering talker Marmoset scenario. The Lombard effect and other effects included are not adequate.

We've added this in the first paragraph of introduction (line 67-69). We've also added a paragraph in discussion (paragraph 2) to compare our results to previous studies.

2. Define "antiphonal calling" for the reader.

We've added a paragraph to detail the antiphonal calling behavior in the method section (line 152-160) and clarified in the result section (line 241-242).

3. Be more specific in defining low and high frequency noise --- does this mimic an interfering call? If not, then what? What specific environmental noises are also being simulated. How accurate are the simulated sounds in reproducing interfering calls and true environemental noises?

Previous studies have described the marmoset habitat noise, which included both low-frequency and high-frequency sounds (Morrill et al., 2013). We have added sentences in the result section to explain that we used the two interfering sounds in our experiments to mimic the noises in the natural environment of marmosets (line 225-228, 235-237). Although these interfering sounds do

not fully capture the spectral profile of the habitat noise, they contain the spectral energy that spans the frequency range of the habitat noise. They were designed to model the acoustic contexts in the natural habitat that can be used to test our hypothesis in this study.

4. The perturbation signal occurs "after the onset of a phee call". Is it overlapping the call and to what extent? Would there be a different effect if not overlapping?

The perturbation signal partially overlapped the phee calls. It started with a delay after vocal onset so it mainly overlapped with the latter half of the phees. If they did not overlap with the ongoing calls, marmosets may choose a strategy to shift the time of vocal production to avoid being masked by the perturbation signals (Roy et al., 2011). We have added explanations of this in the result section (line 251-256).

5. Please describe how the virtual calls and the perturbations are actually implemented with the Marmoset calls in real time. The authors say nothing about this!

Our apology! We have added detailed descriptions in the method section (line 147-175).

6. The statistical analysis appears careful but needs further clarity.

We've added additional text to report detailed statistical analysis in the result section (line 327-329, 380-381).

7. In a real dialogue, would a receiving marmoset not get confused if the caller is anticipating a shift not present and thus would the learning effect stop soon?

The reviewer raised an interesting question here. We cannot address this question by the current study. Our experiment focused on the vocal production of individual marmosets but did not explicitly test whether one marmoset would track the vocal features produced by the other marmoset. Our playback calls for the "virtual conspecific" were selected from a group of pre-recorded phee calls from other marmosets and were presented in random order.

8. In the discussion, authors should tone down the generalizations based on the limited experimental setup.

We have revised the discussion to tone down generalizations based on the reported data. We have also added the following sentence at the end of the first paragraph of the discussion section to indicate the limitation of the present study (line 418-420). In addition, in response to Reviewer-1's critiques, we have carefully re-worded a number of statements throughout the manuscript to make sure we do not overstate the findings of this study (please see our responses to Reviewer-1 above).

"A limitation of the present study is the limited experimental parameters tested due to the challenging nature of these experiments. Future studies shall evaluate the above conclusions in a wider range of acoustic contexts."

9. Figures 1,2 and parts of 3,4 are very confusing in terms of timing of the various signal components. Please clarify.

We have added clarifications in the figure legend.